Author's response to reviews

Title: MIA is a potential biomarker for tumor load in neurofibromatosis type 1

Authors:

Mateusz Kolanczyk (kolanshy@molgen.mpg.de)
Victor Mautner (v.mautner@uke.uni-hamburg.de)
Nadine Kossler (kossler@molgen.mpg.de)
Rosa Nguyen (rosi.nguyen@gmail.com)
Jirko Kühnisch (jirko.kuehnisch@charite.de)
Tomasz Zemojtel (zemojtel@molgen.mpg.de)
Aleksander Jamsheer (jamsheer@wp.pl)
Eike Wegener (wegener.eike@googlemail.com)
Boris Thurisch (mail@boris-thurisch.de)
Sigrid Tinschert (Sigrid.Tinschert@tu-dresden.de)
Nikola Holtkamp (nikola.holtkamp@charite.de)
Su-Jin Park (su-jin.park@charite.de)
Patricia Birch (birch@interchange.ubc.ca)
David Kendler (kendler@ca.inter.net)
Anja Harder (anja.harder@ukmuenster.de)
Stefan Mundlos (stefan.mundlos@charite.de)
Lan Kluwe (kluwe@uke.de)

Version: 9 Date: 14 April 2011

Author's response to reviews:

Authors' contributions: Please indicate that all authors read and approved the final manuscript.

--> Indication was introduced

References: Please make sure references are cited in numerical order. We notice references 19-21 are not cited within the main text. Please make sure references continue to be cited in numerical order once these references are cited.

--> References were once again re-formatted with EndNote (BMC medical education format) and checked. All references are cited in the manuscript body. Ref nr. 9 is positioned throughout the manuscript several times as required. All references are cited in the numerical order.

Figures: The first figure referenced in the manuscript text should be Figure 1. The second, figure 2, etc. Please reorder the figure citation so that they are cited in consecutive order. All figures must be cited; figures not cited will be excluded from the final manuscript. Please do not use ranges when listing figures. Please list all the figures, for example: Figures 5, 6, 7, 8 and 9.
--> all figures are now cited.